Author’s response to reviews

Title: ASSESSING THE EFFICACY AND SAFETY OF MAGNESIUM SULFATE FOR MANAGEMENT OF AUTONOMIC NERVOUS SYSTEM DYSREGULATION IN VIETNAMESE CHILDREN WITH SEVERE HAND FOOT AND MOUTH DISEASE

Authors:

qui phan (phantuqui@gmail.com)
Lam Khanh (lampk@oucru.org)
Truong Khanh (truonghuukhanh@gmail.com)
Trieu Huynh (huynhtrieu82@gmail.com)
Giang Pham (chuotchui1007@yahoo.com)
Bich Nguyen (drngocbich@yahoo.com.vn)
Quyen Tran (bstranthanhquyen@gmail.com)
Vuong Huynh (vuonghuynh_vn@yahoo.com)
Tien Nguyen (ntmytienmdc@yahoo.com)
Thoa le (thoalephan@gmail.com)
Nhan Le (drnhanbvnhidongl@yahoo.com.vn)
Saraswathy Sabanathan (saras.whitehorn@googlemail.com)
Rogier Doorn (rvandoorn@oucru.org)
Tan Le (tanlv@oucru.org)
Toan Nguyen (toannd@oucru.org)
Laura Merson (laura.merson@ndm.ox.ac.uk)
Dung Nguyen (dungntp@oucru.org)
Ronald Geskus (rgeskus@oucru.org)
Hung Nguyen (drthanhhung@gmail.com)
Dear Editors,

Thank you very much for your comments. For comment 1, I have revised the "Availability of Data and Materials" section and updated the link that data can be accessed now. For comment 2, the necessary things have been done. The manuscript and all of figures, table, addition files are clean now. They are also submitted in separated files.

If you have any other comments, Please let me know.

Best Regards,

Dr Phan Tu Qui